Author's response to reviews

Title: An amino acid-electrolyte beverage may increase cellular rehydration relative to carbohydrate-electrolyte and flavored water beverages

Authors:

Chih-Yin Tai (edison@musclepharm.com)
Jordan M. Joy (jordan.joy@spartans.ut.edu)
Paul H. Falcone (paul@musclepharm.com)
Laura R. Carson (laura@musclepharm.com)
Matt M. Mosman (matt.mosman@musclepharm.com)
Justen L. Straight (rdintern@musclepharm.com)
Susie L. Oury (rdintern@musclepharm.com)
Carlos Mendez Jr (rdintern@musclepharm.com)
Nick J. Loveridge (nick.loveridge@musclepharm.com)
Michael P. Kim (doc@musclepharm.com)
Jordan R. Moon (jordan@musclepharm.com)

Version: 3 Date: 29 April 2014

Author's response to reviews:

Our manuscript received revision from three separate qualified reviewers. Few changes to the manuscript were requested. However, we have changed the abstract and methods section so that they are consistent in their reports. Specifically, reviewer found a typo in the descriptive data in one section, and reports of percent dehydration were reported to the tenth in one section yet to the hundredths in another. We have corrected the typo and reduced all reports of dehydration status to the tenths.